Author's response to reviews

Title: Molecular imaging of potential bone metastasis from differentiated thyroid cancer: A case report

Authors:

Nora Sandu MD (norasandu@yahoo.com)
Gabriele Pöpperl MD (g.Poepperl@klinikum-stuttgart.de)
Marie-Elisabeth Toubert MD (marie-elisabeth.toubert@sls.aphp.fr)
Belachew Arasho MD (belachewda@yahoo.com)
Toma Spiriev MD (spiriev@gmail.com)
Mikael Orabi MD (mikael.orabi@lrb.aphp.fr)
Bernhard Schaller MD DSC (bernhardjschaller@gmail.com)

Version: 2 Date: 14 September 2011

Author's response to reviews: see over
Point-by-point Response

We thank the peer-reviewers for their valuable comments that have further improved our manuscript. Please see here in the following our point-by-point response to the peer-reviewer’s comments:

1) Associate Editor
Spelling mistakes and grammatical errors → Done

2) Editorial Team
Include patient’s ethnicity in the abstract and case presentation sections → Done
Include the authors contribution section → Done

3) Peer-Reviewer 2
Abstract, please correct "a few case reports exists" to "a few case reports exist" → Done
Discussion first line: please correct "relative" to "relatively" and "clinical" to "clinically" → Done

4) Peer-Reviewer 1
No suggestion.